Dear Prof. Parkin,

Thank you very much for accepting our revised manuscript (2nd version) [MS: 6971848811237836] for publication in "BMC-Complementary and Alternative Medicine".

Following the suggestion of "The BioMed Central Editorial Production Team", we made the necessary changes (in the text, figure legends and figures) in the revised manuscript (3rd version) as mentioned below:

Figure 1a has been changed as figure 1; figure 1b has been changed as figure 2; figure 2 has been changed as figure 3 and so on...........

Kindly acknowledge the receipt of the revised (3rd version) manuscript. I look forward to see the published manuscript in the journal.

Best regards

Dr. Parames C. Sil
Reader
Department of Chemistry
Bose Institute
93/1, A.P.C. Road
Calcutta-700009
West Bengal, INDIA
Phone : 9133-23506619/2402/2403; ext 412(lab), 413(office)FAX : 9133-2350-6790
E-mail:parames@bosemain.boseinst.ac.in / parames_95@yahoo.co.in